Author's response to reviews

Title: Household out-of-Pocket medical expenditures and National Health Insurance in Taiwan: income and regional inequality

Authors:

Tu-Bin T.B. Chu (benette@wanfang.gov.tw)
Tsai-Ching T.C. Liu (tching@mail.ntpul.edu.tw)
Chin-Shyan C.S. Chen (stan@mail.ntpu.edu.tw)
Yi-Wen Y.W. Tsai (ivytsai@nhri.org.tw)
Wen-Ta W.T. Chiu (wtdhiu@wanfang.gov.tw)

Version: 4 Date: 1 August 2005

Author's response to reviews: see over
Dear Editor, 

Please find attached file containing the revised version of our manuscript entitled “Household out-of-pocket medical expenditures and National Health Insurance in Taiwan: income and regional inequality”

Deeply thank you for your patience. The grant number was released this week finally. Our revised version has been done with all formatting changes requested. By the way, could you please give us a final big help? We wish that our paper can be published after September 1st since our university will provide academic award of US$1000 for the paper published after September 1st (new semester begins). It sounds ridiculous but it is true. We hope that we can get this award and sincerely ask you for another great help. We are really sorry for putting you so much inconvenience. The reviewing procedure of your esteemed journal was too efficient to be believed. This makes us somewhat in the trouble because we did not expect this paper to be published this year according our experiences. We will highly recommend your esteemed journal to my colleagues. We sincerely appreciate your great help and we will not make this mistake again.

Sincerely yours,

____________________
Tu-Bin Chu, MS. MHA.
Vice Superintendent
Taipei Municipal Wan Fang Hospital, Taipei, Taiwan

____________________
Tsai-Ching Liu, Ph.D.
Professor
Department of Public Finance
National Taipei University, Taipei, Taiwan
Researcher
Taipei Municipal Wan Fang Hospital, Taipei, Taiwan

____________________
Chin-Shyan Chen, Ph.D.
Professor
Department of Economics
National Taipei University, Taipei, Taiwan
Researcher
Taipei Municipal Wan Fang Hospital, Taipei, Taiwan

Yi-Win Tsai, Ph.D.
Associate Investigator
Division of Health Policy Research
National Health Research Institutes, Taipei, Taiwan

Wen-Ta Chiu, M.D. Ph.D.
Professor & Vice-President
Taipei Medical University, Taipei, Taiwan
Superintendent
Taipei Municipal Wan Fang Hospital, Taipei, Taiwan